Author's response to reviews

Title: The Right Non-recurrent Laryngeal Nerve with Extralaryngeal bifurcation, The Left Recurrent Laryngeal Nerve also with Extralaryngeal bifurcation and with Sympathetic Nerve Anastomosis: Three Variations of the Laryngeal Nerve in the Same Patient.

Authors:

Emin Gurleyik (egurleyik@yahoo.com)

Version: 3 Date: 15 February 2011

Author's response to reviews:

There are a number of matters that require your attention before your manuscript can be accepted to the Journal of Medical Case Reports:

- Please include the age AND ethnicity of the patient.
- Please remove the sub-headings from the case report section of the manuscript.
- Please include a statement of competing interests.
- Please include a list of abbreviations.
- Please confirm whether the patient provided consent for publication of the images, as well as the case report itself.

Dear Editor

I revised the final version of the manuscript according to these statements. Some revisions and corrections are made in proper sections and at the end of the present report.

Sincerely